Author's response to reviews

Title: The effect of giving influenza vaccination to general practitioners: a controlled trial [NCT00221676]

Authors:

Barbara Michiels (barbara.michiels@pandora.be)
Hilde Philips (philips.hilde@pandora.be)
Samuel Coenen (samuel.coenen@ua.ac.be)
Fernande Yane (f.yane@iph.fgov.be)
Toon Steinhauser (toonkugha@gmail.com)
Sofie Stuyck (sofiestuyck@hotmail.com)
Joke Denekens (joke.denekens@skynet.be)
Paul Van Royen (paul.vanroyen@ua.ac.be)

Version: 4 Date: 15 June 2006

Author's response to reviews:

Dear Editor

I accepted all the changes made to the text without adding new ones.
By using Endnote, corrections were made to the references as asked in the querie.
Yours sincerely,

Barbara Michiels
Corresponding author